Author's response to reviews

Title: Prenatal Exposure to Herbal Remedy in a Girl with Autism Spectrum Disorder: A Case Report

Authors:

Juan A Ortega-García (ortega@pehsu.org)
Mario G Angulo (marioangulofmd15@gmail.com)
Elías J Sobrino-Najul (elias.sobrino@upr.edu)
Offie P Soldin (os35@georgetown.edu)
Alberto Puche-Mira (albertopuche@ono.com)
Eduardo Martinez-Salcedo (edusan@forodigital.es)
Luz Claudio (luclaudio@mssm.edu)

Version: 4 Date: 4 December 2010

Author's response to reviews:

Dear editor,

All formatting changes suggested have been made and all the information required has been included to submission. In the text, only 1 new word we have included to explain the patient’s ethnicity: ‘white’. Also, we provide the full citation details for reference 8.

Thank you,

Ortega García JA, MD, PhD